

© 2022 American Psychological Association ISSN: 0012-1649

2023, Vol. 59, No. 2, 229-235 https://doi.org/10.1037/dev0001421

Meta-Analytic Evidence Against Sex Differences in Infants' and Toddlers' Preference for Prosocial Agents

Francesco Margoni¹, Katharina Block², Kiley Hamlin³, Norbert Zmyj⁴, and Toni Schmader³

¹ Department of Psychology, University of Oslo

² Department of Psychology, University of Amsterdam

³ Department of Psychology, The University of British Columbia

⁴ Institute of Psychology, TU Dortmund University

Can well-documented gender differences in evaluations of prosocial versus antisocial actions found in childhood and adulthood be traced to sex differences in basic sociomoral preferences in infancy? We provide an answer to this question by meta-analyzing sex differences in preference for prosocial over antisocial agents in a set of 53 samples of American and European infants and toddlers aged between 4 and 32 months (N = 1,094). Although the original studies were agnostic to sex differences, we were able to retrieve the original data sets and estimate the effect of infants' and toddlers' sex on sociomoral preferences. Employing both a standard frequentist and a Bayesian approach to meta-analysis, we found strong evidence supporting the absence of sex differences in sociomoral preferences among infants and toddlers. We discuss the relevance of this finding for theories and descriptions of the emergence and developmental trajectory of gender differences in morality.

Public Significance Statement

We found strong evidence that, within the first 3 years of life, males and females do not differ in their tendency to prefer prosocial agents over antisocial agents, a tendency that is considered by many to be a building block of adult moral cognition. The present meta-analysis gives evidence to evaluate theories and hypotheses about the development of gender differences in moral judgment.

Keywords: sex differences, meta-analysis, infancy, morality, prosociality

This article was published Online First September 15, 2022 Francesco Margoni D https://orcid.org/0000-0002-8109-228X

Francesco Margoni and Katharina Block share first authorship and order of authorship was determined by a coin-toss.

This research was funded by Social Sciences and Humanities Research Council of Canada (SSHRC) Grant 430-2018-00361 to Toni Schmader; Natural Sciences and Engineering Research Council of Canada Grant RGPIN-2016-03775 and SSHRC Grant 35-2019-0387 to Kiley Hamlin

We thank Dag-Erik Eilertsen for advice on the statistical analyses.

All data and additional materials are available on the Open Science Framework (https://osf.io/uq3zs/).

Francesco Margoni served as lead for conceptualization, analysis, and writing-review and editing. Katharina Block contributed equally to conceptualization, writing-review and editing, and served in a supporting role for analysis. Kiley Hamlin contributed equally to conceptualization, and served in a supporting role for analysis, supervision, and writingreview and editing. Norbert Zmyj contributed equally to conceptualization, and served in a supporting role for analysis and writing-review and editing. Toni Schmader contributed equally to conceptualization, and served in a supporting role for analysis, supervision, and writing-review and editing.

Correspondence concerning this article should be addressed to Francesco Margoni, Department of Psychology, University of Oslo, Forskningsveien 3A, 0373, Oslo, Norway, or Katharina Block, Department of Psychology, University of Amsterdam, Nieuwe Achtergracht 129B, 1018, Amsterdam, the Netherlands. Email: francesco.margoni@psykologi.uio.no or k.a.e.block@uva.nl

Prosociality is one of the cornerstones of human societies. The cooperation afforded by widespread prosocial behaviors such as helping and sharing is among the main factors that have allowed the human species unprecedented success (Boyd & Richerson, 2009; Henrich & Muthukrishna, 2021). Humans' tendencies to expect prosocial interactions and to evaluate prosocial others positively are key components of cooperation (Krebs, 2008). In adults, valuing prosociality and actively caring for others is linked to greater personal and relationship well-being, suggesting that prosociality is broadly beneficial (Le et al., 2018).

Evidence for the capacity to evaluate prosociality can be found even in preverbal infants. A rich and growing literature documents that children under the age of three years view prosocial agents more favorably than antisocial agents, as indicated by their tendency to preferentially interact with prosocial agents (e.g., Geraci & Surian, 2011; Hamlin et al., 2007). Indeed, a recent meta-analysis including 61 experiments and over 1,200 infants and toddlers aged four to 32 months found that about two thirds of them preferred a prosocial agent (who provided help or distributed resources fairly) over an antisocial one (who hindered or distributed unfairly), and that the tendency to prefer prosocial to antisocial agents was consistent across these first 32 months of life (Margoni & Surian, 2018). Moreover, infants' preference for prosocial agents has been shown to predict better parent-reported social adjustment years later, suggesting that these early judgments do not merely reflect an understanding of moral behavior, but may also tap into children's emerging social functioning in important ways (Tan et al., 2018).

Despite a general tendency for humans to value prosociality, many studies reveal gender differences in prosocial beliefs, preferences, and values in older children and adults. Women appear to be more concerned about prosociality in their moral judgements than men are. Specifically, two meta-analyses indicate that adult women value ethics of care and fairness more than men do, and are more concerned when antisocial acts violate these principles (Atari et al., 2020; Jaffee & Hyde, 2000). A similar gender difference is evident in men's and women's self-reported traits and values: Despite cross-national variation in the size of these reported gender differences, women in most countries report valuing prosocial actions more strongly (Falk & Hermle, 2018; Hsu et al., 2021), and describe themselves with more prosocial traits than do men (Schwartz & Rubel, 2005).

Whereas gender differences in preferences for prosociality are nearly ubiquitous in adulthood, far less is known about the early developmental trajectory of sex or gender differences in prosocial preferences. Of course, one challenge of assessing this question is that the same methods used to document these differences in older children or adults (largely done using self-report measures) cannot be employed with infants and young children. Nonetheless, children as young as six years of age show gender differences in valuing prosocial goals; on average, girls think it is more important to be helpful and kind than boys do (Block et al., 2018; Ojanen et al., 2005). Similarly, by age 10, girls judge antisocial agents more harshly than do boys (Sierksma et al., 2014), and adolescent girls are more likely than adolescent boys to expect story characters to behave prosocially (Eisenberg et al., 2001). While these tasks vary across different age groups (because of constraints of developmental appropriateness), these findings broadly suggest that gender differences in the evaluation of prosocial acts are present by the time children reach elementary school. These studies, however, did not examine children younger than six, leaving the emergence of such gender differences unclear.

The question that guided the current article is whether any aspects of these gender differences in prosocial preferences could be found in even younger age groups. Preverbal infants cannot express the degree to which they value being kind to others, but they can reveal an understanding of, and even a preference for, prosocial over antisocial agents. In fact, this component of early prosociality has been demonstrated in children as young as 3 months old (Hamlin et al., 2010). However, sex differences in these preferences have not been systematically examined.

The onset of sex differences is domain-specific and thus requires an independent investigation for each domain. Indeed, research suggests that there are some sex differences that arise in infancy; for example, by one year of age girls prefer dolls over toy cars, and boys show the opposite bias (Todd et al., 2018). Other beliefs and behaviors, instead, become sex differentiated in middle childhood (Bian et al., 2017; Halim et al., 2011). Thus, whether sex differences specific to the propensity to prefer prosociality over antisociality are found in infancy is an entirely open question requiring further attention and investigation.

Investigating early sex differences in preferences for prosocial versus antisocial agents has the potential to shed light on the origins of sex and gender differences in moral judgment. Indeed, sociomoral preferences in preverbal infants have been shown to be related to subsequent human sociomoral functioning in preschool (Tan et al., 2018); here, observed relationships were only present in boys (note that there were no sex differences in tendencies to show prosocial preferences in infancy in this-quite small-sample). Of course, the earlier sex differences in a construct emerge, the less likely it is that culturally transmitted gender stereotypes or sex-typed parental behavior create these differences. Given this, if sex differences in a construct are observable in very young infants, one could argue that sociocultural learning processes have played a less significant role in producing them, and therefore that hardwired biological factors may also be responsible. For instance, it has previously been claimed that biologically hardwired factors contribute to young girls' higher social interest and empathy relative to boys (Baron-Cohen et al., 2005); these biological factors are undoubtedly reinforced and strengthened by sociocultural norms during development, but these norms may not be solely responsible for gender differences. According to this reasoning, then, it is at least plausible that young girls will prefer prosociality more than young boys do, even in infancy. However, no research to date has had sufficient power to adequately examine this proposition.

Although research has not directly examined sex differences in infants' or toddlers' preference for prosocial over antisocial agents, a few articles, in passing, note the absence of significant sex differences in their samples (Loheide-Niesmann et al., 2021; Salvadori et al., 2015; Tan et al., 2018; Vaish et al., 2010). These individual studies, however, were not adequately statistically powered to assess sex differences. Given this dearth of research, it is unclear when sex differences in preferences for prosociality can be found in infancy, constraining our ability to adjudicate between different explanations for their presence later in life. Yet given how important humans' preference for prosociality is both for societal functioning and personal well-being, it is important to answer this question.

The current work provides a unique critical test of sex differences in one important type of early sociomoral preferences, by taking advantage of a large collection of existing studies on infants' and toddlers' preference for prosocial agents. Specifically, we examined whether, across 53 samples, there is meta-analytic evidence that female (compared with male) infants and toddlers show a stronger tendency to prefer prosocial to antisocial agents.

Method

For this work, we collected additional information of infants within studies of a recent meta-analytic dataset, which assessed the extent to which infants and toddlers prefer prosocial to antisocial agents (Margoni & Surian, 2018). The original meta-analysis included 61 effect sizes (44 published and 17 unpublished). The effect size of interest in Margoni and Surian (2018) was the number of children preferring the prosocial agent divided by the number of children included in the experiment.

The main aim of the present work was to assess whether the sex of the infant or toddler had an effect on the preference for prosocial over antisocial agents. As no sex information was tabulated in the original meta-analyses, we recontacted all the authors of the studies initially included. The studies are described in detail in the online supplemental material on OSF (https://osf.io/uq3zs/). We retrieved a total of k = 53 sex difference effect sizes (of the original 61 from Margoni & Surian, 2018) from single experiments (40 published, 13 unpublished) with a total of 1,094 children (547

female). This final dataset included experiments conducted on children aged between 139 days (4.6 months) and 960 days (32 months), $M_{Age} = 398$, SD = 221 (of these, about 62%, n = 33, were experiments conducted on infants aged ≤ 12 months and 29 days, whereas n = 20 were conducted on older infants and toddlers, allowing us to test for potential developmental effects).¹

The studies we retrieved were all included in the prior meta-analysis (Margoni & Surian, 2018) based on three inclusion criteria: (a) they included a measure for a preference between a prosocial and an antisocial agent, where prosocial agents were either helping, fair, or giving agents, and antisocial agents were either hindering, unfair, or keeping/ taking agents; (b) they tested infants' or toddlers' preferences by specifically employing a manual task (i.e., children were encouraged to manually reach or pick up one of two agents, to selectively help one, or to preferentially give to or receive from the agent some resources); and (c) they included infants or toddlers with a sample-specific average age between four and 36 months.² This age range was selected because four months was the youngest age where children have been reliably documented to be able to express a preference through their reaching behavior (Hamlin, 2014), and including toddlers up to 36 months allowed to assess pivotal hypotheses regarding the presence versus absence of developmental changes in children's sociomoral preferences during infancy and toddlerhood.

Results

To estimate sex differences on infants' and toddlers' sociomoral preferences, we first converted each result into a pair of values: a log odds ratio (OR; expressing the increase in the odds of choosing the prosocial over the antisocial agent if the child is male vs. female) and its associated sampling variance. In the analyses, we used the log OR instead of the OR to respect the assumption of normality required by parametric statistical tests. Next, we used a two-step approach. First, we conducted a standard frequentist meta-analysis of the observed effects to test whether there was evidence of a sex difference. Second, we employed a Bayesian approach to estimate the strength of the evidence in favor of accepting the null hypothesis of no sex difference.

We conducted a random-effects meta-analysis with k = 53 sex differences and a restricted maximum-likelihood estimation for τ^2 (i.e., estimated amount of heterogeneity among the true effects). We used a random-effects approach instead of a fixed-effects approach to take into account both within- and between-study variability, and assumed that the *k* effects are a sample from a larger population (Viechtbauer, 2010).

The estimated average log *OR* was .009, 95% confidence interval (CI) [-.273, .290], z = .06, p = .952. For ease of interpretation, this value can be backtransformed to an *OR*, which is 1.009, 95% CI (.761, 1.336). Thus, the odds of a male child choosing a prosocial agent is 1.009 times the odds of a female child choosing a prosocial agent. In other words, male infants and toddlers were a negligible and statistically nonsignificant .9% more likely to prefer the prosocial agent than were female infants and toddlers (see Figure 1).

Specifying the null hypothesis in this way does not reflect our subjective assessment of the evidence already accumulated in the field in favor of the absence of a sex difference in infants' preferences for prosociality prior conducting this meta-analysis. Rather, this analysis simply allowed us to estimate the positive (Bayesian) evidence in favor of the fact that the data here meta-analyzed are distributed in a way that clearly suggests the absence of an effect of sex (i.e., similarly to a distribution centered on 0)—something that was not a possibility under the standard frequentist approach.

The Bayesian random-effects meta-analysis, with the same k = 53 effects (log *ORs*), supported the null hypothesis of no sex difference in children's preferences (Bayes factor $[BF]_{01} = 11.43$). This number indicates that the observed data are about 11 times more likely to occur under the null hypothesis of no sex difference in sociomoral preferences than under the alternative hypothesis (i.e., a sex difference in preference for prosociality in one of the two possible directions) and is generally taken as indicating strong evidence (Stefan et al., 2019).

No significant heterogeneity between studies was found, Q(52) = 46.80, p = .678. The estimated amount of total heterogeneity τ^2 was .00, 95% CI [.00, .60], and Higgins' I² was 0%, 95% CI [0%, 35%] (Higgins et al., 2003). Thus, the variability of sex differences between data sets was negligible and left little to no heterogeneity to be explained with additional analyses. However, for completeness, in the SM we report a series of analyses testing whether age, sample size, type of scenarios employed, significance of the original preference effect, and other potentially relevant factors moderated sex differences, none of which was significant.

Here, as an example, we report analyses with the variable age as a moderator. The effect of sex on infants' preferences was not significantly moderated by infants' age. This is true both when age was inserted as a continuous variable (average age in days for each experiment), $Q_M(1) = 2.28$, p = .131 (see Figure S2a in the online supplemental materials on OSF), and when it was inserted as a binary variable (young vs. old; where young = $M_{Age} \le 12$ months, 29 days), $Q_M(1) = 2.24$, p = .135. The estimated average *OR* is 1.202, 95% CI [.836, 1.729] for experiments with younger infants (n = 33), and .775, 95% CI [.496, 1.210] for experiments with older infants and toddlers (n = 20).

Last, worrying that the accuracy of the meta-analytic estimates can be hampered by the inclusion of small studies, an additional metaanalysis was conducted selecting the 10% of all the reported effects with the largest sample size (this approach is termed Top10 strategy; Stanley et al., 2010; see also Margoni & Shepperd, 2020). The random-effects meta-analysis with the six isolated effect sizes (approximately 10% of k = 53; see online supplemental materials on OSF for the list of these effect sizes) estimated an average *OR* of 1.009, 95% CI [.588, 1.728], further indicating no evidence for sex differences on children's preferences.

The frequentist approach employed above provided no evidence that the null hypothesis (of no sex difference) should be rejected, but it cannot estimate the strength of the evidence in support of the null hypothesis. To do this, we next employed a Bayesian approach. We conducted this analysis by setting the distribution expressing the null hypothesis of a lack of sex differences as normal and centered on 0, with a standard deviation of $1.5.^3$

¹ These age ranges were the same as in Margoni and Surian (2018) and were chosen because they allowed us to compare two sets of studies similar in size while still being able to compare children in their first year of life with older children.

² To maintain sample homogeneity, Margoni and Surian's (2018) metaanalysis did not include one study that employed preferential looking instead of a manual task (Hamlin et al., 2010).

³ Please also note that whereas it seems clear that the data favor the null hypothesis, the strength of this conclusion hinges substantially on the prior or null hypothesis specification, as is often the case with Bayesian analyses (Gronau et al., 2021).

Author(s) and Year		OR [95% CI]
Dunfeld & Kuhlmeier (2010) Geraci & Surian (2011) A Geraci & Surian (2011) B Hamlin & Wynn (2011) B Hamlin & Wynn (2011) B Hamlin et al (2011) A Hamlin et al (2011) A Hamlin et al (2011) C Hamlin et al (2011) C Hamlin et al (2011) C Hamlin et al (2012) A Scarf et al (2012) A Scarf et al (2012) A Scarf et al (2012) C Dahl et al (2013) A Dahl et al (2013) A Hamlin (2013) A Hamlin (2013) A Hamlin (2013) C Hamlin (2013) C Hamlin (2013) F Hamlin (2015) A Hamlin (2015) A Hamlin (2015) B Hamlin (2015) A Hamlin (2015) A Hamlin (2015) B Scola et al (2015) B Scola et al (2015) B Steckler et al (2017) A Surian & Franchin (2017) A Surian & Franchin (2017) A Surian & Franchin (2017) A Surian & Franchin (2017) A Surian & Sommerville (unpub) B Enright & Sommerville (unpub) B Enright & Sommerville (unpub) D Enright & Sommerville (unpub) F Hamlin (unpub) B Woo & Hamlin (unpub) A		$\begin{array}{c} 0.28 \ [0.02, \ 3.58]\\ 8.75 \ [0.90, \ 84.80]\\ 0.03 \ [0.00, \ 0.74]\\ 33.00 \ [1.43, 760.63]\\ 0.033 \ [0.03, \ 4.19]\\ 1.50 \ [0.07, \ 31.57]\\ 4.20 \ [0.07, \ 31.57]\\ 4.20 \ [0.03, \ 35.12]\\ 1.40 \ [0.14, \ 13.57]\\ 0.31 \ [0.02, \ 4.41]\\ 1.44 \ [0.27, \ 7.83]\\ 2.33 \ [0.17, \ 32.58]\\ 0.24 \ [0.02, \ 3.01]\\ 1.00 \ [0.10, \ 9.61]\\ 0.36 \ [0.05, \ 2.73]\\ 3.00 \ [0.19, \ 47.96]\\ 8.56 \ [0.34, \ 212.94]\\ 5.00 \ [0.17, \ 150.92]\\ 1.00 \ [0.13, \ 7.57]\\ 1.00 \ [0.14, \ 7.10]\\ 11.67 \ [0.46, \ 295.21]\\ 3.33 \ [0.28, \ 40.29]\\ 0.20 \ [0.01, \ 4.91]\\ 0.58 \ [0.04, \ 8.15]\\ 0.14 \ [0.04, \ 4.95]\\ 0.14 \ [0.04, \ 4.95]\\ 0.27 \ [0.02, \ 3.09]\\ 0.70 \ [0.13, \ 3.68]\\ 2.75 \ [0.40, \ 18.88]\\ 1.17 \ [0.24, \ 5.62]\\ 1.38 \ [0.23, \ 8.30]\\ 2.00 \ [0.38, \ 10.41]\\ 0.71 \ [0.04, \ 14.35]\\ 0.94 \ [0.16, \ 5.46]\\ 0.23 \ [0.047, \ 10.35]\\ 0.70 \ [0.24, \ 2.02]\\ 0.60 \ [0.02, \ 3.61]\\ 0.71 \ [0.04, \ 4.76]\\ 0.25 \ [0.047, \ 10.35]\\ 0.70 \ [0.24, \ 2.02]\\ 0.60 \ [0.09, \ 3.99]\\ 0.10 \ [0.00, \ 2.477]\\ 0.30 \ [0.02, \ 3.68]\\ 0.60 \ [0.08, \ 4.76]\\ 0.40 \ [0.06, \ 2.89]\\ 2.89 \ [0.12, \ 7.13]\\ 1.78 \ [0.24, \ 14.77]\\ 2.38 \ [0.05, \ 3.68]\\ 0.60 \ [0.08, \ 4.76]\\ 0.40 \ [0.06, \ 2.89]\\ 2.89 \ [0.12, \ 7.13]\\ 1.78 \ [0.24, \ 14.77]\\ 2.38 \ [0.52, \ 10.48]\\ 0.52 \ 1.68 \ 14.77]\\ 0.31 \ [0.55, \ 1.68]\\ 0.54 \ [0.44, \ 14.77]\\ 0.30 \ [0.05, \ 3.68]\\ 0.60 \ [0.08, \ 4.76]\\ 0.40 \ [0.06, \ 2.89]\\ 2.89 \ [0.12, \ 7.13]\\ 1.78 \ [0.24, \ 14.77]\\ 0.33 \ [0.55, \ 10.48]\\ 0.55 \ 1.68 \ 1.57\\ 0.56 \ 1.5$
Woo & Hamlin (unpub) B Woo & Hamlin (unpub) C		$\begin{array}{cccccccccccccccccccccccccccccccccccc$
RE Model		1.01 [0.76, 1.34]
	0 5 10 15 20	
Average Effect Size = OR		

Figure 1

Forest Plot Displaying Odds Ratios and 95% Confidence Intervals

Note. Forest plot is for the k = 53 studies included in the meta-analysis. Arrows indicate that the upper value of the confidence interval (CI) is greater than 20. An odds ratio (*OR*) >1 indicates that male infants and toddlers were more likely to choose the prosocial agent than female infants and toddlers. The vertical dotted line at 1 represents the reference point indicating no effect of sex on sociomoral preferences. RE = random effects model.

Discussion

Distinguishing and preferring prosocial actions and agents over antisocial actions and agents is one key aspect of sociomoral cognition that contributes to the functioning of human societies (Boyd & Richerson, 2009; Krebs, 2008). In adults, there is a well-documented tendency for women to show greater concern for prosociality across various domains (Atari et al., 2020; Falk & Hermle, 2018; Jaffee & Hyde, 2000; Schwartz & Rubel, 2005) and for girls by age six to express greater value for being kind and helpful (Block et al., 2018). One of the early building blocks of these prosocial values might be infant's tendency to prefer prosocial over antisocial agents (Tan et al., 2018). Whereas infants as young as five months reliably make evaluations leading them to selectively approach prosocial versus antisocial agents (Margoni & Surian, 2018; Van de Vondervoort & Hamlin, 2018), no research to date has adequately tested for sex differences in this phenomenon. Utilizing a large collection of studies in children aged four to 32 months, the current meta-analysis evaluated evidence for sex differences in infants' and toddlers' preference for prosocial agents.

Unlike other studies that have documented early sex differences in a variety of phenomena including toy preferences (Todd et al., 2018) and spatial abilities (Lauer et al., 2019), here we find no evidence for sex differences in infants' and toddlers' preference for prosocial agents. Our findings suggest that both male and female infants and toddlers tend to prefer prosocial over antisocial agents and do so to a similar extent. A large collection of studies and the combination of a traditional frequentist meta-analysis with a Bayesian approach increases the confidence in this null finding. Moreover, because we meta-analyzed both published and unpublished studies that were not originally designed to assess sex differences, publication bias was unlikely to affect our estimates (a finding supported by additional analyses reported in online supplemental materials on OSF).

These results contribute to our understanding of one of the potential building blocks of gender differences in prosociality. Whereas infants' and toddlers' preferences for prosocial agents assessed in tasks used by the meta-analyzed studies are not to be equated fully to the complex moral judgements and actions by older children or adults, these preferences may still be an important basis for prosociality, inasmuch as they have been linked to relatively higher sociomoral functioning later in life (Tan et al., 2018). Adding to what we know about the developmental trajectory of sex differences in morality, the current findings suggest that gender differences in valuing prosociality found in older children (Block et al., 2018; Ojanen et al., 2005; Sierksma et al., 2014) and adults (Atari et al., 2020; Falk & Hermle, 2018; Hsu et al., 2021; Jaffee & Hyde, 2000; Schwartz & Rubel, 2005) do not derive from early sex differences in the very basic ability to evaluate prosocial and antisocial behavior, and to prefer the former to the latter.

Our conclusion that there are no sex differences in infants' and toddlers' sociomoral preferences has several notable limitations. First, the meta-analyzed studies measured infants' and toddlers' perception of prosociality in a single, forced-choice trial. In this literature, the strength or stability of the preference, which could be assessed by testing the same individual using multiple test trials or measures, is currently unknown (but see Nighbor et al., 2017; who suggest these preferences might not be stable). Thus, the fact that the meta-analyzed studies relied on such a coarse measure of preference is a limitation of this work, inasmuch as more subtle sex differences in evaluating prosociality might have been obscured. Indeed, whereas our analysis shows that infants and toddlers of both sexes hold a preference of prosocial agents when provided with a single forced choice, we cannot exclude the possibility that the sexes vary in the strength of this preference. A valuable suggestion for future research, then, would be to develop tasks to assess to what degree infants and toddlers value prosociality, which would provide a more fine-grained test of early sex differences in sociomoral evaluation.

Second, and relatedly, the paradigm of the meta-analyzed studies is unable to distinguish between infants' liking of prosocial characters and their dislike of antisocial characters. Introducing a systematic assessment of infants' and toddlers' preferences for prosocial and antisocial versus neutral agents may also help reaching greater clarity on the nature of these early emerging preferences and, as a result, could allow testing new predictions about the development of gender differences. For instance, it is possible that gender differences found later in childhood or adulthood are related to an early emerging preference for prosocial agents (rather than a dispreference for antisocial ones) that is stronger in girls than in boys.

Both these limitations point to a common and general caveat. The primary goal of this study was solely to assess whether a sex difference in one specific, and quite basic, form of moral cognition—preference for prosocial actors—is present in infancy. Our

findings cannot rule out the possibility that the sex or gender differences in morality found in older children and adults are related to, originate from, and are perhaps explained by, different or more subtle early-emerging sex differences in other processes that were not measured in the studies meta-analyzed here. Finding a sex difference in the extent to which sociomoral preferences are expressed in infants would have suggested that later emerging sex differences in morality may be traced back to infancy. In contrast, our finding of no evidence for a sex difference in this core preference for prosociality could have a variety of implications. A null effect can suggest that later sex or gender differences in prosociality result from social learning. But it is also consistent with the possibility that later observed differences are governed by other earlier emerging sex differences in moral cognitions or preferences that are simply not captured by the tasks employed in the studies we meta-analyzed. Alternatively, these later differences might be the result of inherent sex differences that are not yet observable during infancy but could be supported by biological influences that emerge only later in development (e.g., hormones). In sum, while we show strong evidence for an absence of early sex differences in one type of moral cognition (i.e., the preference for prosocial over antisocial agents), these results merely scratch the surface of the complex task of explaining where and how later emerging sex differences in morality emerge.

Last, the current work might be considered somewhat outdated, as it only includes experiments on infants' preference for prosocial others conducted prior to November 2016. However, there are two reasons to be wary of this concern: first, our results are based on more than 1,000 children and provide strong evidence for the absence of an effect of sex on infants' and toddlers' preferences. Second, we are aware of no work published after 2016 reporting sex differences in children's preference for prosocial agents. Given this, it seems highly unlikely that adding more recent studies would substantially change our effect size estimate.

To conclude, we have analyzed the results of over 50 experiments to assess whether there is evidence of an early emerging sex difference in the proclivity to choose prosocial over antisocial agents, which, as a fundamental sociomoral judgment, is an important precursor of more complex social functioning. The current work provides meta-analytic evidence for the absence of such a sex difference in infants' and toddlers' preference for prosocial others. This result can be taken into account in evaluating the possible pathways to the gender differences in morality found later in childhood and adulthood.

References

- Abramson, L., Dar, M., Te'eni, A., & Knafo-Noam, A. (2016). Preferences for helpers and hinderers in 9- and 18-month-old infants [Unpublished raw data].
- Atari, M., Lai, M. H. C., & Dehghani, M. (2020). Sex differences in moral judgements across 67 countries. *Proceedings. Biological Sciences*, 287(1937), 20201201. https://doi.org/10.1098/rspb.2020.1201
- Baron-Cohen, S., Knickmeyer, R. C., & Belmonte, M. K. (2005). Sex differences in the brain: Implications for explaining autism. *Science*, *310*(5749), 819–823. https://doi.org/10.1126/science.1115455

- Bian, L., Leslie, S. J., & Cimpian, A. (2017). Gender stereotypes about intellectual ability emerge early and influence children's interests. *Science*, 355(6323), 389–391. https://doi.org/10.1126/science.aah6524
- Block, K., Gonzalez, A. M., Schmader, T., & Baron, A. S. (2018). Early gender differences in core values predict anticipated family versus career orientation. *Psychological Science*, 29(9), 1540–1547. https://doi .org/10.1177/0956797618776942
- Boyd, R., & Richerson, P. J. (2009). Culture and the evolution of human cooperation. *Philosophical Transactions of the Royal Society of London. Series B, Biological Sciences*, 364(1533), 3281–3288. https://doi.org/10 .1098/rstb.2009.0134
- Cowell, J. M., & Decety, J. (2015). Precursors to morality in development as a complex interplay between neural, socioenvironmental, and behavioral facets. *Proceedings of the National Academy of Sciences of the United States of America*, 112, 12657–12662. https://doi.org/10.1073/ pnas.1508832112
- Dahl, A., Schuck, R. K., & Campos, J. J. (2013). Do young toddlers act on their social preferences? *Developmental Psychology*, 49, 1964–1970. https://doi.org/10.1037/a0031460
- Dunfield, K. A., & Kuhlmeier, V. A. (2010). Intention-mediated selective helping in infancy. *Psychological Science*, 21, 523–527. https://doi.org/ 10.1177/0956797610364119
- Eisenberg, N., Zhou, Q., & Koller, S. (2001). Brazilian adolescents' prosocial moral judgment and behavior: Relations to sympathy, perspective taking, gender-role orientation, and demographic characteristics. *Child Development*, 72(2), 518–534. https://doi.org/10.1111/1467-8624.00294
- Enright, E. A., & Sommerville, J. A. (2016). Selectivity, consistency, and stability of prosociality in infancy and toddlerhood [Manuscript submitted for publication].
- Falk, A., & Hermle, J. (2018). Relationship of gender differences in preferences to economic development and gender equality. *Science*, 362(6412), eaas9899. https://doi.org/10.1126/science.aas9899
- Franchin, L., Savazzi, F., Neira-Gutierrez, I. C., & Surian, L. (2019). Toddlers map the word 'good' to helping agents, but not to fair distributors. *Journal of Child Language*, 46, 98–110. https://doi.org/10.1017/ s0305000918000351
- Geraci, A., & Surian, L. (2011). The developmental roots of fairness: Infants' reactions to equal and unequal distributions of resources. *Developmental Science*, 14(5), 1012–1020. https://doi.org/10.1111/j.1467-7687.2011.01048.x
- Gronau, Q. F., Heck, D. W., Berkhout, S. W., Haaf, J. M., & Wagenmakers, E. J. (2021). A primer on Bayesian model-averaged meta-analysis. Advances in Methods and Practices in Psychological Science, 4(3), 1–19 https:// doi.org/10.1177/25152459211031256
- Halim, M. L., Ruble, D. N., & Amodio, D. M. (2011). From pink frilly dresses to 'one of the boys': A social-cognitive analysis of gender identity development and gender bias. *Social and Personality Psychology Compass*, 11(11), 933–949. https://doi.org/10.1111/j.1751-9004.2011.00399.x
- Hamlin, J. K. (2013). Failed attempts to help and harm: Intention versus outcome in preverbal infants' social evaluations. *Cognition*, 128, 451–474. https://doi.org/10.1016/j.cognition.2013.04.004
- Hamlin, J. K. (2014). Context-dependent social evaluation in 4.5-month-old human infants: The role of domain-general versus domain-specific processes in the development of social evaluation. *Frontiers in Psychology*, 5, 614. https://doi.org/10.3389/fpsyg.2014.00614
- Hamlin, J. K. (2015). The case for social evaluation in preverbal infants: Gazing toward one's goal drives infants' preferences for helpers over hinderers in the hill paradigm. *Frontiers in Psychology*, 5, Article 1563. https://doi.org/10.3389/fpsyg.2014.01563
- Hamlin, J. K. (2016). A box show replication attempt [Unpublished raw data].
- Hamlin, J. K., Ullman, T., Tenenbaum, J., Goodman, N., & Baker, C. (2013). The mentalistic basis of core social cognition: Experiments in

preverbal infants and a computational model. *Developmental Science*, 16, 209–226. https://doi.org/10.1111/desc.12017

- Hamlin, J. K., & Wynn, K. (2011). Young infants prefer prosocial to antisocial others. *Cognitive Development*, 26, 30–39. https://doi.org/10 .1016/j.cogdev.2010.09.001
- Hamlin, J. K., Wynn, K., & Bloom, P. (2007). Social evaluation by preverbal infants. *Nature*, 450(7169), 557–559. https://doi.org/10.1038/nature06288
- Hamlin, J. K., Wynn, K., & Bloom, P. (2010). Three-month-olds show a negativity bias in their social evaluations. *Developmental Science*, 13(6), 923–929. https://doi.org/10.1111/j.1467-7687.2010.00951.x
- Hamlin, J. K., Wynn, K., Bloom, P., & Mahajan, N. (2011). How infants and toddlers react to antisocial others. *Proceedings of the National Academy of Sciences of the United States of America*, 108, 19931–19936. https://doi.org/10.1073/pnas.1110306108
- Henrich, J., & Muthukrishna, M. (2021). The origins and psychology of human cooperation. *Annual Review of Psychology*, 72(1), 207–240. https://doi.org/10.1146/annurev-psych-081920-042106
- Higgins, J. P., Thompson, S. G., Deeks, J. J., & Altman, D. G. (2003). Measuring inconsistency in meta-analyses. *BMJ*, 327(7414), 557–560. https://doi.org/10.1136/bmj.327.7414.557
- Hsu, N., Badura, K. L., Newman, D. A., & Speach, M. E. P. (2021). Gender, "masculinity," and "femininity": A meta-analytic review of gender differences in agency and communion. *Psychological Bulletin*, 147(10), 987–1011. https://doi.org/10.1037/bul0000343
- Jaffee, S., & Hyde, J. S. (2000). Gender differences in moral orientation: A meta-analysis. *Psychological Bulletin*, 126(5), 703–726. https://doi.org/ 10.1037/0033-2909.126.5.703
- Krebs, D. L. (2008). Morality: An evolutionary account. *Perspectives on Psychological Science*, 3(3), 149–172. https://doi.org/10.1111/j.1745-6924.2008.00072.x
- Lauer, J. E., Yhang, E., & Lourenco, S. F. (2019). The development of gender differences in spatial reasoning: A meta-analytic review. *Psychological Bulletin*, 145(6), 537–565. https://doi.org/10.1037/bul0000191
- Le, B. M., Impett, E. A., Lemay, E. P., Jr., Muise, A., & Tskhay, K. O. (2018). Communal motivation and well-being in interpersonal relationships: An integrative review and meta-analysis. *Psychological Bulletin*, 144(1), 1–25. https://doi.org/10.1037/bul0000133
- Loheide-Niesmann, L., de Lijster, J., Hall, R., van Bakel, H., & Cima, M. (2021). Toddlers' preference for prosocial versus antisocial agents: No associations with empathy or attachment security. *Social Development*, 30(2), 410–427. https://doi.org/10.1111/sode.12487
- Margoni, F., & Shepperd, M. (2020). Changing the logic of replication: A case from infant studies. *Infant Behavior and Development*, 61, 101483. https://doi.org/10.1016/j.infbeh.2020.101483
- Margoni, F., & Surian, L. (2018). Infants' evaluation of prosocial and antisocial agents: A meta-analysis. *Developmental Psychology*, 54(8), 1445–1455. https://doi.org/10.1037/dev0000538
- Nighbor, T., Kohn, C., Normand, M., & Schlinger, H. (2017). Stability of infants' preference for prosocial others: Implications for research based on single-choice paradigms. *PLoS ONE*, *12*(6), e0178818. https://doi .org/10.1371/journal.pone.0178818
- Ojanen, T., Grönroos, M., & Salmivalli, C. (2005). An interpersonal circumplex model of children's social goals: Links with peer-reported behavior and sociometric status. *Developmental Psychology*, 41(5), 699–710. https://doi.org/10.1037/0012-1649.41.5.699
- Salvadori, E., Blazsekova, T., Volein, A., Karap, Z., Tatone, D., Mascaro, O., & Csibra, G. (2015). Probing the strength of infants' preference for helpers over hinderers: Two replication attempts of Hamlin and Wynn (2011). *PLoS ONE*, 10(11), e0140570. https://doi.org/10.1371/journal .pone.0140570
- Scarf, D., Imuta, K., Colombo, M., & Hayne, H. (2012). Social evaluation or simple association? Simple associations may explain moral reasoning in infants. *PLoS ONE*, 7, Article e42698. https://doi.org/10.1371/journal .pone.0042698

- Schwartz, S. H., & Rubel, T. (2005). Sex differences in value priorities: Cross-cultural and multimethod studies. *Journal of Personality and Social Psychology*, 89(6), 1010–1028. https://doi.org/10.1037/0022-3514.89.6.1010
- Scola, C., Holvoet, C., Arciszewski, T., & Picard, D. (2015). Further evidence for infants' preference for prosocial over antisocial behaviors. *Infancy*, 20, 684–692. https://doi.org/10.1111/infa.12095
- Sierksma, J., Thijs, J., Verkuyten, M., & Komter, A. (2014). Children's reasoning about the refusal to help: The role of need, costs, and social perspective taking. *Child Development*, 85(3), 1134–1149. https://doi .org/10.1111/cdev.12195
- Stanley, T. D., Jarrell, S. B., & Doucouliagos, H. (2010). Could it be better to discard 90% of the data? A statistical paradox. *The American Statistician*, 64(1), 70–77. https://doi.org/10.1198/tast.2009.08205
- Steckler, C. M., Woo, B. M., & Hamlin, J. K. (2017). The limits of early social evaluation: 9-month-olds fail to generate social evaluations of individuals who behave inconsistently. *Cognition*, 167, 255–265. https:// doi.org/10.1016/j.cognition.2017.03.018
- Stefan, A. M., Gronau, Q. F., Schönbrodt, F. D., & Wagenmakers, E. J. (2019). A tutorial on Bayes Factor Design Analysis using an informed prior. *Behavior Research Methods*, 51(3), 1042–1058. https://doi.org/10 .3758/s13428-018-01189-8
- Surian, L., & Franchin, L. (2017). Toddlers selectively help fair agents. *Frontiers in Psychology*, 8, Article 944. https://doi.org/10.3389/fpsyg .2017.00944

- Tan, E., Mikami, A. Y., & Hamlin, J. K. (2018). Do infant sociomoral evaluation and action studies predict preschool social and behavioral adjustment? *Journal of Experimental Child Psychology*, 176, 39–54. https://doi.org/10.1016/j.jecp.2018.07.003
- Todd, B. K., Fischer, R. A., Di Costa, S., Roestorf, A., Harbour, K., Hardiman, P., & Barry, J. A. (2018). Sex differences in children's toy preferences: A systematic review, meta-regression, and meta-analysis. *Infant* and Child Development, 27(2), e2064. https://doi.org/10.1002/icd.2064
- Vaish, A., Carpenter, M., & Tomasello, M. (2010). Young children selectively avoid helping people with harmful intentions. *Child Development*, 81(6), 1661–1669. https://doi.org/10.1111/j.1467-8624.2010.01500.x
- Van de Vondervoort, J. W., & Hamlin, J. K. (2018). The early emergence of sociomoral evaluation: Infants prefer prosocial others. *Current Opinion in Psychology*, 20, 77–81. https://doi.org/10.1016/j.copsyc.2017.08.014
- Viechtbauer, W. (2010). Conducting meta-analyses in R with the metafor package. Journal of Statistical Software, 36(3), 1–48. https://doi.org/10 .18637/jss.v036.i03
- Woo, B. M., & Hamlin, K. (2016). Older, but not younger, babies prefer helpers in a cartoon version of the hill paradigm [Manuscript in preparation].

Received January 24, 2022 Revision received April 20, 2022

Accepted May 28, 2022